

ഹ

Earth's Future

RESEARCH ARTICLE

10.1029/2022EF003408

Key Points:

- A model framework for the Rhine basin was developed to simulate streamflow during extreme past drought years in future conditions
- Extreme low flows as in 1976, 2003, and 2018 would aggravate in a future with declined glacier cover and snow pack
- Repeating the drought and heatwave of 2003 in the future results in largest reductions in summer streamflow (70% upstream, 30% downstream)

Supporting Information:

Supporting Information may be found in the online version of this article.

Correspondence to:

M. van Tiel, vantiel@vaw.baug.ethz.ch

Citation:

van Tiel, M., Weiler, M., Freudiger, D., Moretti, G., Kohn, I., Gerlinger, K., & Stahl, K. (2023). Melting alpine water towers aggravate downstream low flows: A stress-test storyline approach. *Earth's Future*, *11*, e2022EF003408. https://doi. org/10.1029/2022EF003408

Received 9 DEC 2022 Accepted 7 MAR 2023

Author Contributions:

Conceptualization: Marit van Tiel, Markus Weiler, Kerstin Stahl Formal analysis: Marit van Tiel Funding acquisition: Markus Weiler, Kerstin Stahl Investigation: Marit van Tiel Methodology: Marit van Tiel, Daphné Freudiger, Irene Kohn Project Administration: Kerstin Stahl Software: Greta Moretti, Kai Gerlinger Writing – original draft: Marit van Tiel

© 2023 The Authors.

This is an open access article under the terms of the Creative Commons Attribution-NonCommercial License, which permits use, distribution and reproduction in any medium, provided the original work is properly cited and is not used for commercial purposes.

Melting Alpine Water Towers Aggravate Downstream Low Flows: A Stress-Test Storyline Approach

Marit van Tiel^{1,2,3}, Markus Weiler⁴, Daphné Freudiger^{1,5}, Greta Moretti⁶, Irene Kohn¹, Kai Gerlinger⁶, and Kerstin Stahl¹

¹Faculty of Environment and Natural Resources, Environmental Hydrological Systems, University of Freiburg, Freiburg, Germany, ²Now at Laboratory of Hydraulics, Hydrology and Glaciology (VAW), ETH Zurich, Zurich, Switzerland, ³Now at Swiss Federal Institute for Forest, Snow and Landscape Research (WSL), Birmensdorf, Switzerland, ⁴Faculty of Environment and Natural Resources, Hydrology, University of Freiburg, Freiburg, Germany, ⁵Department of Geography, University of Zurich, Zurich, Switzerland, ⁶Hydron GmbH, Karlsruhe, Germany

Abstract Droughts can lead to extreme low flow situations in rivers, with resulting severe impacts. Upstream snow and ice melt in many of the world's mountain water towers can alleviate the hydrological consequences of drought, yet global warming threatens the cryosphere. To improve the understanding of melt water contributions during drought in the case of future glacier retreat, we developed stress-test storyline scenarios to model streamflow and tested them in the European river Rhine basin. Meteorological conditions of past drought and low flow years in Europe, 1976, 2003, and 2018, were repeated at three future moments in time, representing nowadays, near future and far future conditions. The latter two conditions were obtained by climate projections under the RCP8.5 scenario. Results show that the low flow situations caused by the meteorological drought situations aggravate in future conditions, more so for the far future and for the year 2003 because of the relatively large glacier ice melt contribution in the past. Summer (July-September) streamflow may decline by 5%-25% far downstream and 30%-70% upstream and the duration of extreme low flow situations may double compared to the selected past drought events. These results are relevant for the Rhine as a major European river but stand exemplary for many other river basins and highlight the importance of cryospheric changes for downstream low flow situations in a changing climate. The stress-test scenarios allow a glimpse into future extreme low flow events aiding adaptation planning, and might be adapted to include other important low flow drivers.

Plain Language Summary Extended periods with strongly reduced rainfall, in combination with hot summers, lead to accumulating water shortages. As a result, water levels in rivers drop which causes problems, e.g., for shipping, cooling of power plants and drinking and irrigation water supply. During such drought periods, melt water from snow and ice is important for water supply. However, glaciers are projected to further decline in a warming climate, possibly worsen future low flow situations. To quantify this effect, we modeled the amount of water flowing through the Rhine basin (a) for past low flow events in 1976, 2003, and 2018 and (b) for hypothetical situations where we repeat the weather data of those past low flow years at three moments in the future. The results show that flows upstream and downstream in the river Rhine would get even lower in future conditions and cause low flow situations to lengthen considerably. Especially for the year 2003, which had high ice melt contributions in the past, changes are large. In summer, the flow during already critical low flow situations may decrease by up to 70% upstream, and by up to 30% downstream. The results show a glimpse into future low flow events and may help adaptation planning.

1. Introduction

Extreme low flow situations, caused by meteorological droughts and heatwaves and their consequent development into hydrological droughts, lead to a range of negative impacts (e.g., Mosley, 2015; Stahl, Kohn, et al., 2016; Van Slobbe et al., 2016; Wan et al., 2021; Yevenes et al., 2018). During such extreme drought situations, melt water from snow and glaciers in the alpine parts of a basin, also called mountain water towers (Immerzeel et al., 2020; Viviroli & Weingartner, 2004), sustains streamflow in the upstream parts and in the streams and rivers flowing further downstream. During (late) summer drought, other runoff generating processes, such as snowmelt or rainfall, are absent or reduced and this causes the relative contribution of glacier melt water to streamflow to increase, thus amplifying its importance, even far downstream (Ayala et al., 2020; Huss, 2011;



Writing – review & editing: Marit van Tiel, Markus Weiler, Daphné Freudiger, Greta Moretti, Irene Kohn, Kai Gerlinger, Kerstin Stahl Kaser et al., 2010; Pritchard, 2019; Stahl, Weiler, et al., 2016). In addition, when drought situations co-occur with heatwaves, glaciers supply extra melt water that can partly compensate for the lack of precipitation (Pelto et al., 2022; Van Tiel et al., 2021; Zappa & Kan, 2007). During past drought events in western Europe (e.g., in 1976 and 2003), upstream melt in, e.g., the river Rhine basin, could sustain streamflow further downstream, but despite the melt water contributions, these extreme drought events still led to restrictions on water use, impairment of shipping, problems with power plant cooling systems and a deterioration of water quality (e.g., Jonkeren et al., 2007; Massarutto et al., 2013; Vinke et al., 2022).

However, these important melt water contributions during drought situations are under threat due to ongoing and future climate warming. The reduction in melt water supply may worsen critical low flow situations in basins around the world. Climate warming is affecting the cryosphere, with less precipitation falling as snow, earlier start of the snowmelt season and retreating glaciers due to (strong) negative glacier mass balances (e.g., Huss et al., 2017; Nepal, 2016; Nie et al., 2021; Schmucki et al., 2015; Sommer et al., 2020). For the European Alps, Zekollari et al. (2019) showed that glacier volume is expected to reduce by around 50% in 2050 compared to the ice volume in 2017 and by 2/3 to complete disappearance by the end of the 21st century, depending on the future emission scenario. At the same time, the frequency and intensity of extreme meteorological conditions such as droughts and heatwaves are projected to increase in a warming climate (e.g., Beniston et al., 2007; Gobiet et al., 2014; Kundzewicz et al., 2006; Van der Wiel et al., 2021) and current extremes are sometimes presented as a precursor of future events (e.g., Beniston, 2004; Samaniego et al., 2018). How these two processes, future retreating glaciers and extreme droughts, will affect downstream low flows and related water use restrictions in large river basins has not yet been adequately quantified.

The typical model chain used for hydrological climate change impact assessments is forcing a hydrological model with an ensemble of climate model projections and comparing past and future simulations of streamflow. For example, Lutz et al. (2014) used this approach for river basins in High Mountain Asia, Schnorbus et al. (2014) for catchments in British Columbia and Feyen and Dankers (2009) for Europe, with a special focus on low flows. Due to the large spread in future climate projections and the many intertwined hydrological processes taking place, changes in certain streamflow characteristics, e.g., hydrological regime or low and high flow statistics, are commonly analyzed with large uncertainties (Addor et al., 2014; Marx et al., 2017). There has been less attention on analyzing (a) the detailed hydrological processes and (b) specific water use restrictions of future extreme events. This may in part be due to the nature of climate model projections that are statistical representations of future climate model and wet and dry years can vary greatly between different climate models that represent the same future climatic conditions. Extreme years may therefore occur at various moments in the future in different climate model simulations. This may be problematic for analyzing hydrological systems that change over time, such as glacier-fed basins under climate change. The timing of an extreme meteorological event is namely key, because the hydrological system state (e.g., level of glacier retreat) can considerably determine the hydrological response.

Few studies focused on changing streamflow in large glacier-fed river basins and on the role of glaciers during drought in these basins. Pritchard (2019), e.g., calculated the amount of melt water released by glaciers in High Mountain Asia during an average and during a drought year and compared it with precipitation estimates to derive melt fractions for a large set of upstream glacierized dam catchments. The study shows that fractions increase during drought, mainly due to the large decreases in precipitation input rather than increased melt. However, the study did not look at streamflow and as a result did not look at how melt water inputs affected streamflow further downstream. Biemans et al. (2019) simulated the contribution of snow and glacier melt to discharge and irrigation in the Asian basins of the Indus, Ganges, and Brahmaputra, thus coupling upstream flow to downstream water demand. However, the study did not look at drought years in particular. In general, most basin scale glacio-hydrological studies focus either on the upstream parts of a basin, or even solely the glaciers, and provide only a rough estimation of the downstream hydrological impacts of changing flows and melt water contributions (e.g., Comeau et al., 2009; Huss, 2011; Huss & Hock, 2018; Junghans et al., 2011), or they focus on the whole basin but with uncertainties for the upstream alpine parts because of a too coarse resolution and absence of a specific glacier melt and retreat module (e.g., Buitink et al., 2021; Rottler et al., 2021; Shrestha et al., 2012).

For usable predictions, there is a need to apply a modeling framework that includes both the complex alpine parts and the downstream parts of large river basins to assess how streamflow is affected by drought and a changing cryosphere. We suggest to use so called event-based story line scenarios (Shepherd et al., 2018; Sillmann et al., 2021), as a complement to climate model projections, to analyze how known past events are influenced by future cryospheric and basin conditions. This approach overcomes the problem of random timing of future extreme drought events in climate projections. Such alternative modeling scenarios have been used in other contexts and studies, all aiming for (a) a better understanding of process changes and (b) system response related to practice and management during extreme conditions. Builtink et al. (2021), e.g., swapped the model forcing between two different decades, i.e., temperature, precipitation, and evaporation, each in a separate hydrological model run, to disentangle their effects on streamflow changes in the Rhine basin. Stoelzle et al. (2020) and Hellwig et al. (2021) introduced a "stress-test" approach, in which past drought events were modeled and altered by reducing the groundwater recharge before the event for German catchments. They aimed to test the sensitivity of the groundwater system to address relevant questions for water management, namely, "could the event have been even worse?" In the context of changing glacier melt contribution, Koboltschnig et al. (2007) asked the question, "what if the climatically extreme summer of 2003 had happened in 1979?" and applied it as a stress-test storyline. They modeled the extreme summer of 2003 with both the glacier outline of 2003 and the larger one from 1979, for an Austrian headwater catchment and found streamflow amounts would have been up to 12% higher in July and August if the 2003 meteorological conditions had happened in earlier years. "Stresstest storylines," here defined as model scenarios in which extreme events (stresses) from the past are used or are worsened ("stressed"), provide complementary scenarios to general climate scenarios to increase our insights into the sensitivity of the studied hydrological system and its water management.

To analyze the effect of a changing cryosphere on summer flows and downstream low flows during drought in a large river basin, we apply such a stress-test storyline model approach and take the major transboundary European river Rhine basin as an example. As a transboundary organization, the International Commission for the Hydrology of Rhine basin raised interest in these changes and this study builds on models and data of a project to support adaptation planning (Stahl et al., 2022). Here, we focus on the "what-if" question "what will happen if meteorological conditions similar to a past observed low flow situation were to occur again in the future?" We develop a modeling framework to simulate streamflow and its components ice, snow, and rain for the glacierized headwaters and the rest of the Rhine basin. We select three extreme drought years that have occurred in the past, 1976, 2003, and 2018, and compare the streamflow simulations of these original extreme years with the years repeated in different future conditions when glaciers have further retreated and quantify the changes in streamflow. To add value to those model predictions and create directly usable storylines, we also apply critical water use restriction thresholds. These thresholds concern environmental flows, hydropower generation, river navigation, and drought warning levels.

2. Study Area

The Rhine basin was chosen here as a large river basin that originates in a glacierized mountain area, to study the effect of changed glacier and basin conditions on streamflow during extreme past droughts. The river Rhine is one of Europe's major rivers and is of high economic, ecological, and societal importance. It originates in the European Alps in Switzerland and flows through Germany to the Netherlands, with tributaries in France, Austria, Belgium, and Luxembourg (Figure 1). The Rhine basin upstream of Lobith (border between Netherlands and Germany) covers 159,896 km². In 1973, the total glacierized area in the Rhine basin was around 400 km², and covered 0.27% of the basin. The glacier area has been decreasing since then, with 0.21% coverage in 2003 and 0.18% in 2010 (based on data by: Fischer et al. (2014), Müller et al. (1977), Maisch (2000), and Paul et al. (2011)).

To simulate streamflow and its components snow (Q_{snow}) , ice (Q_{ice}) , and rain (Q_{rain}) , the Rhine basin was subdivided into two parts: the glacierized headwaters upstream of the highest gauging stations or first major confluence and the rest of the basin (Figure 1). The glacierized headwaters of the Rhine basin (n = 66), delineated by Freudiger et al. (2020), cover 3% of the Rhine basin upstream of Lobith and vary in size from 10 to 230 km² and have a glacier cover ranging from almost 0% to 38% (Figure 1). Important gauges along the Rhine are the gauge at Basel, that integrates the flow in the upper basin, including the Alps (Hänggi & Weingartner, 2011; Pfister et al., 2006), the gauge at Kaub, that is located at one of the narrowest stretches and thus represents a critical threshold for shipping (Jonkeren et al., 2011), and the gauge at Lobith, that measures all the water that enters the Netherlands and is therefore a critical indicator for drought management plans in the Netherlands (De Vries et al., 2021).





Figure 1. Map of the Rhine basin. The gauges that were used in this study are indicated.

The hydrological regime along the Rhine changes from upstream to downstream. In the upstream parts, flows are high from May to September, due to the melt of snow and ice, and very low during winter when precipitation falls as snow. This pattern continues downstream, but with an earlier timing of the annual peak flow and higher flow during winter. Downstream of Basel, the regime changes with highest flows in winter, and lowest flows in September/October. The precipitation regime also differs within the basin. In most of the basin, summer receives the highest amount of precipitation, but in the eastern part of the basin and downstream of Lobith, in the Netherlands, autumn and winter are wettest. The driest season is spring in the downstream half of the basin, and mostly winter in the upstream half of the basin. Differences in precipitation amounts between the wettest and driest season are higher (>200 mm) in the higher elevated regions of the basin.

3. Methods and Data

3.1. Hydrological Modeling

Daily streamflow and its components Q_{ice} , Q_{snow} , and Q_{rain} for the Rhine basin were modeled using two linked hydrological models, similar as in Stahl, Weiler, et al. (2016) and Stahl et al. (2022). We used a modeling framework to simulate future basin conditions, to apply the stress-test scenarios and to analyze process changes by looking at the different streamflow components.

For the glacierized headwater catchments, the HBV-light model was used (Seibert & Vis, 2012; Seibert et al., 2018). HBV-light has been used in many other studies to simulate streamflow of alpine catchments (e.g.,

Alvarez-Garreton et al., 2021; Finger et al., 2015; Girons Lopez et al., 2020; Konz & Seibert, 2010; Van Tiel et al., 2018). It is a semidistributed model using elevation (in this study at intervals of 100 m and for the glacier 10 m) and aspect classes (three classes) and it includes the delta-*h* parametrization to simulate glacier retreat (Huss et al., 2010; Seibert et al., 2018) and a snow redistribution module, essential in high elevation catchments (Freudiger et al., 2017). The simulated streamflow and its components at the outlet of the headwater catchments were used as lateral inflow to the river network of the hydrological water balance model LARSIM (Large Area Runoff Simulation Model; Ludwig & Bremicker, 2006). The distributed LARSIM model was used to simulate the other tributaries in the Rhine basin and to route the headwater streamflow through the basin. The LARSIM model was used at 1×1 -km resolution upstream of Basel (LARSIM-CH), and 5×5 -km downstream of Basel (LARSIM-ME). The model setup of LARSIM-CH consists of four models and was provided by the "Landesanstalt für Umwelt Baden-Württemberg" and the "Amt der Vorarlberger Landesregierung." The LARSIM-ME model was provided by the "Bundesanstalt für Gewässerkunde" (Germany).

Snow and ice melt is simulated using a degree day approach, with different melt factors for ice and snow in HBV, a degree day approach in LARSIM-ME and an energy balance method in LARSIM-CH. The models furthermore include interception, evaporation, and soil and groundwater flow routines. Potential evapotranspiration (PET) was estimated based on temperature using Hamon for the headwaters. For the LARSIM part of the modeling, actual evapotranspiration was estimated using Penman-Monteith. Reservoirs present in the upper part of the Rhine basin are represented as four cumulative reservoirs for which operation rules have been estimated in the LARSIM model. To track the different streamflow components (Q_{ice}, Q_{snow} , and Q_{rain}) through the basin, a parallel mixing tank with a small volume is used. This way the effect of ice melt, snowmelt, or rainfall input, passing through the storages, lakes, and reservoirs is simulated, rather than the movement of the particles (for a detailed description see: Stahl et al. (2017) and Weiler et al. (2018)). Note that Q_{snow} includes both snowmelt on and off the glaciers.

The HBV-light model was calibrated for each of the headwater catchments on glacier volume change between 1973 and 2010 from modeled ice thicknesses (Fischer et al., 2014; Huss & Farinotti, 2012; Müller et al., 1977), daily snowline elevation from Moderate Resolution Imaging Spectroradiometer (MODIS) snow cover data (Fugger, 2018; Krajčí et al., 2014), a combination of observed and modeled catchment daily Snow Water Equivalent (SWE) between 1,900 and 2,500 m a.s.l. obtained from SLF (Swiss institute for Snow and Avalanche Research), and also streamflow, if available (15/66 catchments) (Freudiger et al., 2020; Van Tiel et al., 2022b). For the ungauged catchments, a regionalization procedure, using all Swiss gauged glacierized catchments (also in Rhone, Ticino, and Inn basins), determined the rainfall-runoff parameters, while snow and ice related parameters were recalibrated (Freudiger et al., 2020). LARSIM was used with an existing parametrization from earlier applications of the model (Stahl, Weiler, et al., 2016).

Input for the HBV-light model consisted of daily precipitation (P) and temperature (T) data, derived from the MeteoSwiss gridded products RhiresD and TabsD ($\sim 2 \times 2$ -km resolution) (MeteoSwiss, 2017, 2019). For LARSIM, the meteorological input, including P, T, radiation, relative humidity, and wind speed, were derived from station data (1,482 stations) and the gridded HYRAS product (Rauthe et al., 2013; Razafimaharo et al., 2020). For the period 2016–2019, a smaller set of stations was used for the interpolation in the LARSIM model due to changes in data availability and gap filling algorithms. To simulate future glacier retreat and streamflow, seven GCM-RCM climate model combinations from the EURO-CORDEX (Jacob et al., 2014) ensemble's RCP8.5 scenario runs (see Figure S4 in Supporting Information S1) were used as model forcing, after a multivariate, including all respective input variables, bias correction had been applied based on the period 1981–2010 (Cannon, 2018; Meyer et al., 2019). The RCP8.5 scenario is used here as a benchmark to derive future conditions and compare the stress-test simulations with. The RCP8.5 scenario, however, only provides one possible future evolution of the climate, and is also called the worst-case scenario (Hausfather & Peters, 2020). We chose only one scenario, since the focus is on understanding the changes rather than attributing them to specific climate scenarios. The RCP8.5 scenario was chosen here as it provides the most pronounced changes and the largest range of future outcomes (i.e., a situation with strong glacier retreat). These are needed to understand the changes in the hydrological mountain river basin system and consequently, identify future risks for water management.

This modeling framework for the Rhine was run in transient mode with Run 1: from October 1973 to 2019, using meteorological observations as input, and with Run 2: from October 1973 to 2100 with the seven GCM-RCM combinations. The starting date of the simulations was determined by the available ice thickness data, needed for

initialization of the model. From Run 1, the streamflow signal during the selected drought years was extracted, to serve as a reference for the stress-test runs. From both Run 1 and Run 2, model states were saved to be used as changed initial future conditions for the stress-test modeling (see Section 3.3).

3.2. Selection of Past Drought Years

Three major drought and low flow years within the observation period 1974–2019 were selected for the stresstest model approach, namely 1976, 2003, and 2018. At the downstream gauging stations in the basin (Basel and Lobith), these 3 years had both the lowest observed 7-day minimum flow in summer and lowest total average summer flow within the selected period. The 3 years cover low flow and extreme drought years in different decades. Associated meteorological conditions of these events have been described in detail in other studies (Baker et al., 2021; Bakke et al., 2020; Buras et al., 2020; Hannaford et al., 2011; Rebetez et al., 2006; Van der Wiel et al., 2021).

Although all 3 years experienced a concurrent drought and heatwave, the 2003 and 2018 low flow years were warmer than 1976, especially in late summer (August and September) and the temperatures were exceptional for a longer period (Figure 2). Precipitation in 1976 was far below average from February to June, and in August. Also, the antecedent year, 1975, was characterized by low precipitation amounts in October and December. In 2003 and 2018, precipitation was below average from February to September, and in 2018, precipitation continued to be below average until November. In contrast to the antecedent conditions of 1976, the last months of 2002 and 2017 were relatively wet, especially November 2002.

3.3. Stress-Test Storylines: Past Drought Years in Future Conditions

To "stress-test" how changed future conditions and glacier retreat affect the streamflow response to extreme meteorological conditions, the model input data from the three drought calendar years (1976, 2003, and 2018) were extracted and used as model forcing (drought-scenarios) at three different moments in time (a) nowadays (2018), (b) near future (2031), and (c) far future (2071). For each of them, the model state at the end of the calendar years 2017, 2030, and 2070, respectively, was saved (Table 1). The saved model state included the glacier extent, the snow, soil, and groundwater storages and the lake and reservoir levels. For the near and far future seven model states, from the different climate model combinations of RCP8.5, were saved (Run 2) and all of them were combined with each of the three drought-scenario years. For the drought-scenario 2018, the nowadays run corresponded with the reference run.

In addition to these initial condition stress-tests, we modeled the selected drought years during their original year of occurrence without the glacier melt contribution (no-glacier scenario) (Table 1). For this, several options were tested. Removing the glacier in the headwaters modeling, i.e., assigning the glacier Hydrological Response Units (HRUs) as nonglacierized, resulted in unrealistic high snow packs due to the missing conversion from snow to ice on the glacier surface. Instead, the $Q_{\rm ice}$ component in the headwaters was subtracted from total streamflow from the model output for the headwaters for the no-glacier scenario. For the rest of the Rhine basin, the changed input from the headwaters (i.e., no-glacier scenario) was routed through the Rhine basin using a separate model run. Due to the reduced amounts of water coming from the upstream headwaters in the no-glacier scenario and reservoir operations in the upstream part of the Rhine basin, changes in the other streamflow components $Q_{\rm snow}$ and $Q_{\rm rain}$ amounts can occur, in otherwise exactly similar conditions as the original model runs.

3.4. Changing Drought, Low Flow, and Water Restriction Characteristics

Climate model data characterizes future occurrences of meteorological drought. To put the meteorological conditions of the stress-test year in perspective of these future projections, they were compared with the RCP8.5 climate model ensemble. For each of the selected drought years, characteristic months were selected regarding precipitation (low) and temperature (high). The characteristic months varied per variable and per drought year. The anomaly of the aggregated characteristic months was calculated based on the hindcast period (1974–2019) for the drought years. The anomaly was calculated for the upstream and the downstream meteorological stations. Here, upstream was defined as the hydrological catchment of the Rhine until Basel with a buffer of 30 km surrounding it, and downstream included the other stations. For these analyses, the selection was based on 915

Earth's Future





Figure 2. Hydrometeorological conditions of the selected drought and low flow years. The first row shows precipitation anomalies including October-December of the previous calendar year. Temperature anomalies and streamflow are shown for the calendar year. The distribution in the precipitation and temperature anomalies refers to the different meteorological stations in the Rhine basin. Streamflow is smoothed with a 7-day moving window.

precipitation stations and 360 temperature stations that had data throughout the period 1974–2019. The upstream part included 195 precipitation stations and 157 temperature stations and the downstream precipitation and 203 temperature stations. The anomalies were based on the mean of the respective station selections. For the future projections, the anomalies were calculated in a similar way, but using the climate model projections of 1974–2019 as a reference. To stress the importance of timing of future extremes, an 11-year period around the "now," "near future," and "far future" conditions were selected. Thus, past conditions (*P* and *T* anomalies) were compared with future meteorological conditions in an 11-year period around the "now," "near future," and "far future" conditions of the climate model ensemble.

Table 1

Overview of the Model Simulation Details of the Different Stress-Test Scenarios

o ver new of the fiload bullman on D entries for the D affertant of						
	Reference	No-glacier	Nowadays	Near future	Far future	
Drought-scenario 1976	Hindcast 1976	Hindcast 1976 without Q_{ice}	IC = 31-12-2017 (Run 1) + met = 1976	IC = 31-12-2030 (Run 2) + met = 1976	IC = 31-12-2070 (Run 2) + met = 1976	
Drought-scenario 2003	Hindcast 2003	Hindcast 2003 without Q_{ice}	IC = 31-12-2017 (Run 1) + met = 2003	IC = 31-12-2030 (Run 2) + met = 2003	IC = 31-12-2070 (Run 2) + met = 2003	
Drought-scenario 2018	Hindcast 2018	Hindcast 2018 without Q_{ice}	x	IC = 31-12-2030 (Run 2) + met = 2018	IC = 31-12-2030 (Run 2) + met = 2018	

Note. Rows indicate the different drought years and the columns represent the different scenarios that were compared. IC = initial condition and met = meteorological forcing, respectively.

The resulting streamflow projections of the stress-test simulations were compared with the original simulations of the past drought years to infer changes in low flows. Results are mostly shown for the gauging stations Weisse Luetschine, Basel, and Lobith, representing a glacierized headwater catchment, the upstream part of the Rhine basin and the downstream outflow, respectively. For the near and far future conditions, the range of simulations shown is coming from the seven climate model forced initial conditions. These results are either represented as different lines or as bars, with the median indicated with a horizontal line. To investigate how extreme the stress-test streamflow simulations are with respect to the recent past, 7-day moving streamflow percentiles were calculated based on the simulated period 1974–2019 (Run 1). Changes in total summer streamflow (July-September) were compared with the ice melt contribution during summer to analyze how changes in flow are related to a reduction in the Q_{ice} contribution to streamflow. To examine the impact of the stress-tests on streamflow at a weekly resolution, calendar weeks with the highest relative ice melt contribution were selected, for each drought year and for each gauging station. This week was then compared with the same calendar week in the stress-test simulations and changes in the three streamflow components were examined.

Four drought and water use thresholds were calculated to analyze how the changed flow conditions in the stresstest simulations affect drought impacts. For the upstream station Bruegg-Aegerten along the Aare, the Q_{347} , or the 95th percentile of the daily streamflow for a reference period, in our study 1980-2010, was calculated. This metric is used in Switzerland to determine minimal flow requirements (Kohn et al., 2019). Bruegg-Aegerten has a flow regime with lowest flows in February and March and the value calculated for the reference period mostly refers to this period. Q_{347} is calculated for the hindcast (Run 1) and for the seven members of the climate model projections. Subsequently, the days with flow lower than Q_{347} are counted per calendar year. The same procedure is done for the various stress-test years. For Basel, the water availability for hydropower production at the Kemps power plant was estimated. For this, the flow above monthly varying minimum flow requirements (52-180 m³/s) and flow needed for the fish pass (8 m³/s) and below the maximum flow limit (1,400 m³/s), was calculated and summed for each calendar year. For Kaub, the days per year that the flow is above the high $(3,445 \text{ m}^3/\text{s})$ or low flow (784 m³/s) shipping thresholds were calculated (GIQ, 2014). For Lobith, the days with drought alert level 1 were calculated. The Rhine flow at Lobith into the Netherlands provides an important water supply and low flows in the Rhine are part of the warning system to alert for drought conditions (De Vries et al., 2021). The threshold for the Rhine varies from 1,000 m³/s in winter and spring, to 1,400 m³/s for May, 1,300 m³/s for June, 1,200 m³/s for July, and 1,100 m3/s for August.

4. Results

4.1. Characteristics of Past Low Flow Years

The different dynamics of the cumulative precipitation deficits and heatwaves in the three drought years resulted in different streamflow responses (Figure 2). In general, the streamflow simulations match the observations during these extreme years well. For 1976 and 2003, the simulations and observations have the same dynamics, but the simulations have a slight positive bias at Basel and Lobith. In 2018, the streamflow simulations are underestimated in the first half year at Basel, but closely follow the observations during the rest of the year and at Weisse Luetschine and Lobith (Figure 2).

In 1976, streamflow in the downstream reaches of the Rhine basin was low throughout the year, in particular from February until the beginning of July. In Lobith (basin outlet), the lowest flow was recorded on July 11 in 1976.

In the model simulations (Run 1), the lowest flow occurred on September 1, which resembles a second minimum flow day also in the observations. According to the model simulations, the snowmelt contribution was average to low, especially in May and June. The ice melt contribution in 1976 was shifted to approximately 1 month earlier compared to the long-term mean and was the lowest compared to the other two drought years. Maximum simulated daily ice melt contributions in 1976 were 39.4%, 5.6%, and 3.2%, at Weisse Luetschine, Basel, and Lobith, respectively.

In 2003, in contrast, the ice melt contribution was much higher than the long-term average, throughout the summer. As a result, the summer streamflow anomalies were positive in 2003 for the upstream gauging station Brienzwiler and for the headwaters that had a relative glacier cover larger than 8%. Flows in 2003 were relatively high between April and June in the headwaters and in the upstream part of the Rhine basin at Basel because of the high snowmelt contribution. Flows were extremely low from July to September at the gauges Basel and Lobith (Figure 2). The lowest flow in Lobith was recorded on September 28 (September 27 in the simulations). Maximum simulated daily ice melt contributions in 2003 were 63.7%, 18.6%, and 12.1%, at Weisse Luetschine, Basel, and Lobith, respectively.

In 2018, the flows in July, August, October, and November at Basel and Lobith were the lowest within the observed study period 1974–2019. In 2018, flows were closer to average conditions until the mid of June, partly due to the above-average snowmelt contribution. Compared to 2003, the low flow situation prolonged beyond September, resulting in the day of lowest flow occurring later in autumn (1 December/25 October at Basel, for observations and simulations, respectively and 29 October/28 October at Lobith. The difference between observations and simulations is also here caused by two low flow minima per year). Maximum daily simulated ice melt contributions in 2018 were 47.8%, 10.9%, and 6.4%, at Weisse Luetschine, Basel, and Lobith, respectively.

4.2. Future Cryospheric and Meteorological Conditions

The simulations forced by the ensemble of future climate scenarios under the RCP8.5 scenario (Run 2) show strong glacier retreat in the Rhine basin and its subcatchments (Figure 3). Compared to the start of the simulations in 1973, the glacier area in the whole Rhine basin has diminished by 36% in 2018. By 2030 and 2070, the glacier area is projected to have declined by 39%–51% and 85%–96%, respectively. The rate of glacier retreat varies between catchments and regions. For some catchments, substantial changes in glacier area are simulated between 1976 and 2003, but for other catchments glaciers retreated more strongly after 2003. Comparing the hindcast simulation, from which the "nowadays" conditions are extracted, with the ensemble of climate projections, from which "near" and "far future" conditions are taken, shows that the hindcast simulation is at the lower end of the climate model ensemble. This may result in a higher glacier cover in some of the "near future" conditions when comparing them with the 2018 year and "now" conditions. For the "far future" conditions, glaciers have strongly retreated and cover an area less than 60 km² in the Rhine basin.

Past extreme drought years have been suggested to be a precursor of future events that may occur more regularly (e.g., Beniston, 2004; Samaniego et al., 2018). To put the stress-test results into perspective, we compared some of the meteorological characteristics of the past drought years with meteorological conditions in the future climate model ensemble (Figure 4). In general, this comparison shows that future extreme drought events are not necessarily much drier, but they are much warmer in the RCP8.5 ensemble, especially for the upstream part of the Rhine basin. Looking at the distributions of future combined precipitation and temperature anomalies, the past selected drought years seem still plausible in a future climate, despite the strong warming trend. However, the results shown here may indicate that not all climate scenarios may potentially capture extreme meteorological drought events (strong P deficits), especially in the downstream parts of the Rhine basin. Overall, the results suggest that meteorological conditions similar to past drought years will occur more frequently under the RCP8.5 scenario.

4.3. Streamflow and Low Flows During Past Drought Years in Future Basin Conditions

If the meteorological conditions of past low flows were to occur again in the future when glaciers will have progressively retreated, low flows aggravate (Figure 5). The relative changes in streamflow between the original drought years and the stress-tested drought-scenario years are larger for the upstream headwater catchment Weisse Luetschine catchment (13.5% glacier cover in 2010), in which glacier and snowmelt contributions have a relatively larger share in total streamflow, compared to the downstream stations Basel and Lobith.





Figure 3. Glacier area evolution in the Rhine basin, its subbasins and in the glacierized headwaters. From 1976 to 2019, the hindcast simulations are shown and from 2020 to 2100, the ensemble of climate model projections of RCP8.5 are shown.

Furthermore, the stress-test effect depends on the drought year. In comparison to 1976 and 2018, the drought-scenario year 2003 shows the largest changes relative to the original low flow year. This can be attributed to the larger ice melt contribution in 2003 because of the exceptionally high temperatures, especially in August, when glacier melt usually peaks (Figure 2). This ice melt component is greatly reduced in the different future conditions stress-tests, due to glacier retreat. The stress-test results show that in the future, flows under drought-scenario 2003 may become extremely low and even out of range of the current lowest flow situations (Figure 5). For drought-scenario years 1976 and 2018, the relative changes in the stress-test scenarios are smaller but are evident around end of June/beginning of July for 1976 and in August for 2018.

The changes in streamflow in the stress-test simulations compared to the reference drought years are larger in the far future conditions compared to the nowadays situation. Nonetheless, according to the simulations, the effect of retreated glaciers would already have been noticeable if the meteorological conditions of 2003 had happened again around the year 2018 (nowadays). For the drought-scenario year 1976 repeated in nowadays conditions, the simulated flow is relatively higher compared to the reference year, for Basel and Lobith (Figure 5). This increase relates to a higher snowmelt component, because the snow pack at the end of 1975 (year preceding the original





Figure 4. Thermopluviograms of the characteristic meteorological conditions of the three drought-scenario years (columns) for upstream (first row) and downstream (second row) areas of the Rhine basin. For each drought-scenario, the anomalies of P and T are based on different months (see Section 3.4) and are indicated with labels upper-left (P anomaly) and lower-right corners (T anomaly). Anomalies of P and T relate to the mean of 1974–2019 observations or climate model data.

year) was lower than in 2017 (year preceding the nowadays conditions). The effect of more snow in winter on streamflow diminishes around midsummer. For the near future and far future conditions, the seven different climate model projections caused slightly different initial conditions and, as a result, small differences in the relative streamflow changes.

The stress-test simulations do not only allow to investigate how past drought years would look like in future conditions but also how each years' meteorological drought conditions influence the hydrological response. Comparing the "nowadays" runs of drought-scenario years 1976 and 2003, with the streamflow of the reference drought year 2018, keeps the antecedent and catchment and glacier conditions similar, and only the meteorological forcing differs. The high temperatures in drought-scenario 2003 in June and August, caused a higher ice melt contribution compared to the other two drought-scenario years (in nowadays and reference conditions, for 1976 and 2018, respectively), even when glaciers have the same area (Figure S1 in Supporting Information S1). The development of Q_{rain} and Q is similar for the drought-scenario years 2003 and 2018 from around June to October. Q_{snow} is generally higher for the drought-scenario years 1976 and 2018 and can likely be attributed to a combination of high precipitation anomalies in January, and colder temperatures in March in those years compared to drought-scenario years flows occur for the conditions of the year 2018, because of the prolonged precipitation deficit. In the headwaters, the close to normal and cold temperature anomalies in September and October in 2003 may have caused the lowest flow of late summer to have occurred in the year 2003, compared to the conditions of the drought-scenario years of 1976 and 2018.

The comparison of the changed future initial conditions stress-tests with the no-glacier scenario shows that the no-glacier scenario is not always the worst-case scenario (Figures 5, 6, and S2 in Supporting Information S1). This shows that not only changes in ice melt contributions are responsible for the change in streamflow but also changes in the snowpack and antecedent catchment storage. For the Weisse Luetschine headwater catchment, removing the ice component of the streamflow lowers the streamflow almost as much as shifting the





Figure 5. Streamflow during selected drought years (columns) in different conditions: nowadays, near future and far future for three gauging stations along the Rhine (rows). The black line shows the original simulated low flow year. For the near and far future, the seven lines correspond to the seven climate model simulations used as forcing to obtain the initial model states. The streamflow time series are smoothed with a 7-day moving window. Note: logarithmic *y*-axis.

low flow year to far future conditions. In contrast, for the downstream stations Basel and Lobith, especially for the drought-scenario 2003, the no-glacier stress-test streamflow decreases less than the three initial conditions stress-tests (Figures 5 and 6). These results can be explained by different processes. First, the relative ice melt contribution is smaller downstream. Second, in the reservoirs downstream of the headwaters but upstream of





Figure 6. Changes in summer streamflow in the stress-test scenarios for the three selected drought years. The left column shows the different gauging station in the Rhine downstream of the glacierized headwaters, and the right column shows the glacierized headwaters. The changes in summer streamflow for "now," "near future," and "far future" conditions are shown against the relative ice melt contribution to streamflow during the original drought year. The black line shows the 1:1 relation between changes in summer streamflow and the relative ice melt contribution.

Basel, the outflow changes because of the changed inflow (less water because of no Q_{ice}). The snowmelt and rain component are higher in the no-glacier scenario compared to the original run and compensating the missing Q_{ice} component, despite having otherwise the same meteorological and antecedent conditions. Third, besides changes in the ice melt contribution to streamflow, the different initial condition stress-test scenarios also affected the drought years' antecedent conditions of groundwater and snow storage, thereby affecting also the other streamflow components, as opposed to the no-glacier scenario in which antecedent conditions did not change.

In general, the relative changes in streamflow in the stress-test scenarios can be related to the catchments' glacier cover and the Q_{ice} contribution. Thus, the streamflow changes decrease from upstream to downstream (Figure 6). Changes in mean July-September streamflow during low flow years in different future conditions were largest for the year 2003. For the highly glacierized headwater catchments, streamflow reduces up to 40% in the summer when examining drought-scenario 1976 in the far future, and up to 70% and 60%, respectively, for drought-scenarios 2003 and 2018 in the far future. If the 1976 and 2003 meteorological conditions occur in the nowadays situation, flows are simulated to be up to 10%–30% lower. Downstream of the headwaters, changes in summer flow are largest for Brienzwiler in the Aare river (Figures 1 and 6). In Basel, changes are around 5%–15%





Figure 7. Flow composition of the calendar weekly maximum relative ice melt contribution for the three drought years, in different scenarios and for different gauges. Vertical red lines indicate the average flow of the selected weeks, based on the simulations 1974–2019.

when repeating the drought-scenario years 1976 and 2018 but are larger for the drought-scenario year 2003 (15%–30%) in future conditions (Figure 6). Downstream, at Lobith, the changes are small for the 1976 drought-scenario year but become substantial for drought-scenario 2003, with 15%–25% change, and drought-scenario 2018, with few percent up to 20% change.

Zooming in at a higher temporal resolution, Figure 7 shows that the relative ice melt contribution in the week with the highest contribution, decreases. While weekly flows were extremely low for Basel and Lobith during the three drought years, upstream melt could compensate for the lack of precipitation. In far future conditions, however, flows are reduced to half and well below current normal conditions upstream. Figure 7 shows that all streamflow components change in the stress-test scenarios, but the largest changes occur in Q_{ice} . Weekly ice melt contributions between 3% and 18% for the gauging stations Basel and Lobith during drought years decrease to negligible contributions in the far future.

4.4. Water Use Restrictions During Stress-Test Scenarios

Some of the changes in flow in the stress-test scenarios shown above may seem small, in particular for downstream gauges along the Rhine. However, these changes in already critical low flow situations may have large





Figure 8. Annual water availability and water use restrictions at several locations along the Rhine: low flow indicator of Switzerland at Bruegg-Aegerten, hydropower production at Basel, Shipping threshold at Kaub and Drought alert level at Lobith. The black dots represent the hindcast simulations, with the three selected drought years indicated in red (reference), and with different symbols. The drought-scenario years in future conditions are indicated with the green, purple, and orange symbols. The blue dots and the blue shaded area show the climate model projections of RCP8.5. The blue line shows the average of the 15 years smoothed seven climate projections and the shaded area the minimum and maximum range.

implications. Figure 8 shows water availability and water use restrictions for different stations along the Rhine and how these change in the stress-test scenarios. At Bruegg-Aegerten, a gauging station in the Aare tributary, 1976 was the year with the longest duration of flow below the Q_{347} threshold (around 100 days), only two other years had low flows for longer, namely 2011 and 2017. In future conditions, the number of days below this low flow threshold increases for all drought-scenario years. The metric is strongly related to flows from January to March in the current conditions, and may eventually also include flows in late summer when flows drop considerably in future conditions and increase in winter. For these reasons, the drought-scenario year 1976 stands out as flows were low in particular during the start of the year 1976. A bit further downstream, at Basel, impacts of drought situations can be felt in terms of hydropower production (Figure 8). The three drought years were among the years with the lowest water availability for hydropower production, with 1976 as the most extreme year. In future conditions, the total flow diminishes and represent future extremes in low water availability for hydropower production. At Kaub, the impaired navigation period was highest for 2018 and negligible in 1976 (Figure 8). In the stress-test scenarios, the impaired navigation period for the drought-scenario years 2003 and 2018 increases with around 50 days from the reference to the far future conditions. Drought-scenario 2003 in near future conditions would make the year the second extreme year, after 2018, with respect to the hindcast period from 1974 to 2019. All the way downstream, at Lobith, 2018 was also the year with the longest duration of the drought alert level (Figure 8). For drought-scenario 1976, the duration of flow below the alert thresholds decreases in the nowadays and most of the near future conditions, but for the other two drought-scenario years the duration increases, more so for the year 2003 than 2018 as the original year already had an exceptional long duration of flows below the threshold.

5. Discussion

5.1. Meltwater Contribution to Downstream Streamflow and Impacts on Low Flows

This study quantified the "what-if"-question "what would happen if past low flow years were to happen again in the future when glaciers have retreated?" The simulations demonstrate that past extreme summer low flow situations worsen in future conditions. How much streamflow changes and the hydrological drought situation aggravates depends on (a) the location in the basin, (b) the timing of reoccurrence, and (c) the specific meteorological conditions.

The relative decreases in summer streamflow during drought years in different future conditions were found to increase with glacier melt contribution to streamflow, which in turn largely depends on the catchments' relative glacier cover. Thus, in general, changes in streamflow in the stress-test simulations were smaller further downstream. However, the flow regimes of glacierized headwater catchments and their downstream basins are very different, with the summer season usually changing from a high to a low flow season from upstream to downstream. Thus, despite smaller relative changes in streamflow, impacts on water availability may be particular profound in the downstream reaches where flow is already critically low during drought summers. The changes in drought characteristics along the river Rhine confirm these patterns, where changes in summer flow around the order of 10%–20% (Figure 6) result in 50% increase in duration of impaired navigation period, e.g. (Figure 8).

The second aspect that determines the exacerbation of past drought years is the timing of reoccurrence. The simulations show that strongest changes are too be expected in far future conditions, when both glaciers in the basin have retreated the most and initial conditions are likely to be drier than nowadays conditions, due to the extreme warming trend. For the drought-scenario years 1976 and 2003, the now and near future conditions do not differ much in terms of glacier area in the basin and the streamflow of these scenarios are therefore close together (Figure 3). The same is true for the reference 2018 drought year and the drought-scenario year 2018 in near future conditions, as they are closer together in time, compared to the reference drought year 2003 and the drought-scenario year 2003 in near future conditions. These years thus have a different benchmark. If similar meteorological conditions as in 1976, 2003, and 2018 occur in the near future, streamflow will already change considerably.

The third control of streamflow changes in the stress-test scenarios is the specific meteorological conditions of the selected low flow years and the specific antecedent storage conditions. The meteorological conditions determine when during the season the lowest flows occur and how much ice could melt during summer, and thus how susceptible streamflow is to changed glacier conditions. In our selected years, the drought year 1976 was mainly characterized by large precipitation deficits, in particular during the winter/spring season, while in the drought year 2003 the extreme heatwave played a role and caused high glacier melt contributions. Antecedent conditions influence the snow and groundwater storage in the catchment, and a large difference in these antecedent conditions between the original and the stress-test conditions results in stronger changes. The results of switching the antecedent year to 2017 instead of 1975 show that if 1976 was not preceded by a dry start of winter in 1975, flows may have been higher. When comparing the years 2003 and 2018, that were both characterized by a heatwave in summer, 2018 turns out to be more extreme, especially when looking at drought characteristics (Figure 8). The reason could be potentially attributed to a decrease in glacier area and glacier melt contribution between 2003 and 2018 (e.g., similar to the case studied by Koboltschnig et al. (2007)), but this can only be tested in a model framework as presented here. Figure S1 in Supporting Information S1 shows that if 2003 and 2018 would occur with the same conditions (glacier area and catchment conditions), flows in Basel and Lobith would have been very similar between June and September, but lower in October and November under drought-scenario 2018 and lower from end of January to beginning of May under drought-scenario 2003 conditions. Q_{icc} was higher under 2003 scenario conditions than under 2018 scenario conditions and the other way around for Q_{snow} . This implies that not only the retreat of the glaciers played a role in the extremeness of the 2018 drought, but also the timing and magnitude of the rainfall anomalies and the summer temperature anomalies.

Overall, the results emphasize that the melt contributions to streamflow diminish in the future and thereby worsen extreme summer low flows downstream in large basins, such as shown here for the Rhine basin. These findings agree with studies from Gosling et al. (2017), Hurkmans et al. (2010), and Marx et al. (2017), that show that low flows will decrease in the Rhine basin under future climate change. However, unlike in our stress-test storylines, these modeling studies did not include the process of glacier retreat and the model setup did not allow to disentangle the role of a changing cryosphere from changing future meteorological conditions.

The findings presented here for the Rhine basin might be expected similar or even stronger in other basins. Whereas the Rhine basin is situated in a relatively wet climate and has a relatively small proportion of glaciers, other basins in Chile and High Mountain Asia, e.g., can contain large ice masses in their mountain water towers and have drier climates, both of which increases the importance of glacier melt for downstream streamflow dynamics (Immerzeel et al., 2020; Kaser et al., 2010; Sorg et al., 2012). Depending on the rate of glacier retreat, the precipitation anomalies and drought conditions and the trend and variability of basin storages, streamflow changes during drought years in future conditions may be more pronounced and potentially leading to large water availability issues (McCarthy et al., 2022; Pritchard, 2019).

5.2. Design of Stress-Test Scenarios

Stress-test scenario's within the framework of event-based storyline approaches have been used in various disciplines and studies to complement probabilistic climate model based projections (e.g., Shepherd et al., 2018; Stoelzle et al., 2020). While the rationale behind different stress-test studies is similar, i.e., aiming for an event-oriented characterization of future risk, implementations vary. To the best of our knowledge, this is the first study that uses this approach in a glacio-hydrological context. Koboltschnig et al. (2007) implemented a similar approach for a glacierized catchment in the Austrian Alps, but the focus was not on risk assessment. Here, we showed the importance of stress-test scenarios for glacio-hydrological systems undergoing rapid changes to understand changes in future risks of extreme low flows and their water use implications. Using only climate model ensembles, the information of specific timing of extreme events in combination with specific glacier states is lost, but important to understand the role of glacier retreat for future low flow risk in large mountain river basins.

Regarding the current design of our stress-test scenarios, a few methodological and process-based considerations can be discussed to aid future design of stress-test scenarios in mountain river basins. The first aspect is the selection window of the meteorological conditions for the stress-test scenarios. Here, the calendar year was chosen. Other selection windows were deemed to be less objective, as, e.g., starting dates of water years vary from upstream to downstream, and different drought events are characterized by different lead-up times. Depending on the importance of snow for summer streamflow, different methodological choices on the drought selection window could lead to slightly different estimates of how streamflow may change in changing conditions. Several studies have shown that winter snow may be important for late summer flows (Godsey et al., 2014; Jenicek et al., 2016; Tague & Grant, 2009), therefore future studies may focus on developing stress-test scenarios that specifically address the question of how changed snow conditions in the winter may affect the development of low flows over the summer from upstream to downstream.

A second aspect is the likelihood of past meteorological conditions in a future climate with a strong warming trend. The results in Figure 4 show that especially toward the end of the century, meteorological conditions can be much warmer than today in the RCP8.5 scenario. However, finding analogs of past drought years in future climate data can be a challenge, because aggregated anomalies do not necessarily need to have the same effect on the hydrological system response and larger ensembles may be needed to test the frequency and characteristics of future extreme drought years (e.g., Brunner et al., 2019; Van der Wiel et al., 2021). Despite this challenge, the warming trend in extreme scenarios (RCP8.5) is evident and future stress-test scenarios should incorporate more extreme (= hotter) droughts and analyze their effect on future low flow risk (Rasmijn et al., 2018; Teuling, 2018; Udall & Overpeck, 2017).

We chose here to obtain the future stress-test conditions from a small climate model ensemble driven by the RCP8.5 scenario and from past conditions (for the "now" situation). In some event-based storyline approaches, "what-if" questions and scenarios can also be designed using imaginary conditions or events, but related to reality (e.g., stakeholder interaction, large ensemble simulations), to investigate extreme rare events. One example in our system could have been to combine an extreme dry winter with an extreme dry summer. However, it is important that physical constrains are not violated. This means that we discourage, e.g., to use changed glacier outlines only to analyze the effect of glacier retreat without changing the catchment conditions that go along with the conditions that lead to glacier retreat.

5.3. No-Glacier Scenario

As a potential "worst-case" scenario, the different drought years were also run in a "no-glacier" scenario. This scenario does, however, not resonate with the above-mentioned physical constrains that should not be violated,

and was therefore also difficult to implement. The results show that compared to the other changed conditions scenarios, only removing the Q_{ice} contribution does not result in the strongest changes. These results and the implementation of the "no-glacier" scenario may have several implications for other studies.

As described before, designing the "no-glacier" scenario was not straightforward, as removing a model glacier where it climatically is supposed to exist, results in undesired model behavior. The large snowpack that builds up in that case stores large amounts of meltwater and rainfall and may affect the simulated streamflow in unintended ways, depending on the way processes are conceptualized in the glacio-hydrological model used. However, the "removing-glacier" technique is used in some model studies, to circumvent the need for tracking the glacier contribution through the system (Weiler et al., 2018). In these cases, flow is compared between the run including glaciers and the run without glaciers (e.g., Comeau et al., 2009; Frans et al., 2018; Jost et al., 2012), potentially wrongly attributing the differences in flow to glacier melt contribution only if the snow-to-ice conversion is not taken into account. Other techniques that are used to answer the "what-if" question of completely disappearance of glaciers, relate to the imbalance component of glacier melt (e.g., McCarthy et al., 2022). However, this potentially misses the subtraction of the part of balanced glacier melt that will not be available in case of glacier disappearance and may be important for late summer streamflow.

In our case, the Q_{ice} part of the flow was removed, without removing the glacier in the model configurations. In terms of processes and flow routing, this has a few consequences. Rainfall on glacierized areas contributes to the glacier liquid storage part and contributes directly to streamflow based on outflow parameters that are dependent on the snow on the glacier. However, rainfall on nonglacierized areas infiltrates to the soil and groundwater storages and may contribute to streamflow with a delay. This process also partly explains the changes in Q_{rain} in the future conditions stress-test scenarios. Overall, modeling the processes of no-glacier landscape as a scenario for current climate conditions does not work and we need to be careful discussing such scenarios, unless such a scenario is combined with a future climate and catchment state, in which glaciers do not exist anymore.

5.4. Model Uncertainties

The modeling framework presented here allowed for an integrated assessment of the changing role of upstream glacier and snowmelt for downstream low flows in large river basins. The Rhine basin is a relatively data-rich basin, as opposed to basins that source in, e.g., High Mountain Asia (Lutz et al., 2014; Nepal, 2016). The data availability of streamflow observations in particular is important to constrain and validate the simulations and volumes of water, besides the cryospheric remote sensing data sets that are becoming more widely and globally available.

Modeling streamflow of a large mountain river basin such as the Rhine basin here brings a number of challenges. Streamflow simulations did not always match very well with observations. However, in large basins this can have a number of causes because of the many processes taking place in the basin at the same time. In our simulations, e.g., the year 1976 fitted less good to the observations in Lobith (overestimation), but the other years were simulated well here (Figure 2). In Basel, streamflow was underestimated in the first half year of 2018 but fitted well with observations for the rest of the year. Although it is important to improve modeling efforts, also in these challenging large scale mountain river basins under extreme drought situations, individual tuning of parameters for different parts of the basin was not applied to avoid overfitting of the model. Rather, we aimed for a consistent modeling framework that allows to analyze plausible changes in streamflow under different stress-test scenarios.

The presented model framework divides streamflow in its components Q_{ice} , Q_{snow} , and Q_{rain} . Validating these contributions with observations downstream in the Rhine basin is impossible, as they do not represent particles but effect tracking contributions (Weiler et al., 2018). The estimations can be compared with other studies in the Rhine basin on glacier melt contributions, Junghans et al. (2011) and Huss (2011) e.g., estimated the glacier storage change contribution in August 2003 to streamflow at Basel to be around 23%–31% and at Lobith 15%. These estimates are slightly higher than our estimates (15% and 8.4%, respectively), but we only look at the ice melt contribution, thus excluding snow on the glacier. Moreover, in our approach, daily melt is simulated and tracked through the hydrological system, whereas the other two studies directly compared glacier runoff with downstream streamflow observations.

A specific aspect that needs to be addressed in studies on future low flow situations downstream of mountain water towers is human interactions with the water system, of which reservoir management is one example. Our results showed that with less Q_{ice} water supply from the headwaters in the case of the "no-glacier" scenario the



outflow of the other streamflow components changes, but based on reservoir operations determined in the past. With future changing streamflow patterns and extremes, reservoir operating rules may change as well. In addition, they may be used to mitigate some of the low flow risk as outlined here in this study if enough water is available during other periods (e.g., Ehsani et al., 2017). Other important facets for model improvement are including changes in vegetation patterns and dynamics (e.g., Duethmann et al., 2020) and multiyear subsurface catchment storage changes (e.g., Fowler et al., 2020), that are currently not well represented in many hydrological models.

6. Conclusion

This study quantified the "what-if"-question "what would happen if meteorological conditions similar to a past observed low flow situations were to occur again in the future?" using a stress-test storyline modeling framework with a special focus on the glacierized headwater catchments of the European river Rhine basin. The simulations demonstrate that past extreme summer low flow situations exacerbate in future conditions. How much they aggravate depends on the location in the basin, the timing of reoccurrence and the specific meteorological conditions of past drought events.

The results emphasize that melt contributions from glaciers and snow to streamflow diminish in the future with glacier retreat and thereby worsen extreme summer low flows downstream in large basins, such as shown here for the Rhine basin. Summer streamflow would reduce by a few percent up to 70% along the Rhine, depending on the gauging station distance from the glaciers, the reference drought year and the timing of reoccurrence. In the headwaters, changes are generally larger. However, changes in summer streamflow do not linearly relate to the Q_{ice} melt contribution, because other processes such as changed initial conditions (groundwater storage and snow) play a role too and vary from catchment to catchment. These findings, together with the results of the no-glacier scenario, warn for simply omitting glaciers in glacio-hydrological simulations to investigate the role of glacier retreat on downstream streamflow dynamics.

Several drought indicators calculated for Basel, Kaub, Lobith, and Bruegg-Aegerten point toward a strong aggravation of hydrological drought impacts, as a result of decrease in summer flow, further exacerbating low flow situations. Water availability for hydropower production would reduce, the impaired navigation period at Kaub may lengthen by up to 2 months and drought alert levels in the Netherlands could double in length, impacting water use. Comparing the drought characteristics of the past years with the future climate model ensemble of RCP8.5, shows that drought impacts as we know them from the years 1976, 2003, and 2018 will occur more often and will be prolonged and more severe.

The stress-test storyline framework enabled both the investigation of future conditions, regarding glacier retreat and catchment states, as well as disentangling the effect of past drought year meteorological and catchment conditions on hydrological drought propagation. In further work, such a framework could be used for attribution of extreme future events with regards to catchment and glacier conditions versus meteorological conditions. As indicated by several studies, past drought events may show a preview of future events that will occur with greater frequency. Thus, the stress-test storylines presented here show a glimpse into future hydrological drought responses that are likely to occur more often and aid communication to stakeholders, which may improve future drought management plans or adaptation strategies. However, droughts are likely to become hotter and the question that remains open is how future droughts will influence the hydrological response and low flows in a mountain water tower context, where glaciers could possibly partly compensate for evaporation and lack of rainfall due to increased melt but are also strongly retreating.

Data Availability Statement

Meteorological station data were obtained from Deutscher Wetterdienst DWD, Bundesamt für Meteorologie und Klimatologie Schweiz, MeteoSchweiz, eHYD (Hydrographischer Dienst und Bundesministerium für Nachhaltigkeit und Tourismus) archive for Austrian data, Koninklijk Meteorologisch Instituut Belgie (KMI), MétéoFrance, DREAL Grand Est and the European Climate Assessment Dataset (ECAD) (Klein Tank et al., 2002). Gridded interpolated meteorological observations for the glacierized headwater catchments were extracted from the TabsD (MeteoSwiss, 2017) and the RhiresD (MeteoSwiss, 2019) products from MeteoSwiss. Climate model data were obtained from EURO-CORDEX, through the ESGF portals of DKRZ and SMHI. Streamflow data were provided by Federal Office for the Environment (FOEN) Switzerland and the Swiss Cantons (Freudiger et al., 2020) and the



Acknowledgments

This work was funded by the CHR, International Commission for the Hydrology of the Rhine basin. We thank Andreas Hänsler for the bias correction of the climate model data and Matthias Huss for providing the ice thickness data. MVT also acknowledges funding from the STAY! Scholarship New University Endowment Freiburg. Open Access funding enabled and organized by Projekt DEAL.

References

Addor, N., Rössler, O., Köplin, N., Huss, M., Weingartner, R., & Seibert, J. (2014). Robust changes and sources of uncertainty in the projected hydrological regimes of Swiss catchments. *Water Resources Research*, 50, 7541–7562. https://doi.org/10.1002/2013WR014979

simulations for the rest of the Rhine basin (Gerlinger et al., 2023) can be found on FreiDok too.

German Federal Institute of Hydrology (BFG) and can be requested from them. Glacier outlines were obtained through Fischer et al. (2014), Müller et al. (1977), Maisch (2000), Paul et al. (2011). The HBV-light model can be downloaded from https://www.geo.uzh.ch/en/units/h2k/Services/HBV-Model/HBV-Download.html. The stress-test model simulations are available on FreiDok https://freidok.uni-freiburg.de/ (Van Tiel et al., 2023). The reference simulations for the glacierized headwaters (Van Tiel et al., 2022a, 2022b), as well as the reference

- Alvarez-Garreton, C., Boisier, J. P., Garreaud, R., Seibert, J., & Vis, M. (2021). Progressive water deficits during multiyear droughts in basins with long hydrological memory in Chile. *Hydrology and Earth System Sciences*, 25(1), 429–446. https://doi.org/10.5194/hess-25-429-2021
 Ayala, L., Farías-Barahona, D., Huss, M., Pellicciotti, F., McPhee, J., & Farinotti, D. (2020). Glacier runoff variations since 1955 in the Maipo
 - river basin, semiarid Andes of central Chile. The Cryosphere, 14(6), 2005–2027. https://doi.org/10.5194/tc-14-2005-2020
- Baker, L., Shaffrey, L., & Hawkins, E. (2021). Has the risk of a 1976 north-west European summer drought and heatwave event increased since the 1970s because of climate change? *Quarterly Journal of the Royal Meteorological Society*, 147(741), 4143–4162. https://doi.org/10.1002/ qj.4172
- Bakke, S., Ionita, M., & Tallaksen, L. (2020). The 2018 northern European hydrological drought and its drivers in a historical perspective. *Hydrology and Earth System Sciences*, 24, 5621–5653. https://doi.org/10.5194/hess-2020-239
- Beniston, M. (2004). The 2003 heat wave in Europe: A shape of things to come? An analysis based on Swiss climatological data and model simulations. *Geophysical Research Letters*, 31, L02202. https://doi.org/10.1029/2003GL018857
- Beniston, M., Stephenson, D. B., Christensen, O. B., Ferro, C. A., Frei, C., Goyette, S., et al. (2007). Future extreme events in European climate: An exploration of regional climate model projections. *Climatic Change*, 81(Suppl. 1), 71–95. https://doi.org/10.1007/s10584-006-9226-z
- Biemans, H., Siderius, C., Lutz, A. F., Nepal, S., Ahmad, B., Hassan, T., et al. (2019). Importance of snow and glacier meltwater for agriculture on the Indo-Gangetic Plain. *Nature Sustainability*, 2(7), 594–601. https://doi.org/10.1038/s41893-019-0305-3
- Brunner, M. I., Farinotti, D., Zekollari, H., Huss, M., & Zappa, M. (2019). Future shifts in extreme flow regimes in Alpine regions. *Hydrology and Earth System Sciences*, 23(11), 4471–4489. https://doi.org/10.5194/hess-23-4471-2019
- Buitink, J., Melsen, L. A., & Teuling, A. J. (2021). Seasonal discharge response to temperature-driven changes in evaporation and snow processes in the Rhine Basin. *Earth System Dynamics*, 12(2), 387–400. https://doi.org/10.5194/esd-12-387-2021
- Buras, A., Rammig, A., & Zang, C. S. (2020). Quantifying impacts of the 2018 drought on European ecosystems in comparison to 2003. *Biogeosciences*, 17, 1655–1672. https://doi.org/10.5194/bg-17-1655-2020
- Cannon, A. J. (2018). Multivariate quantile mapping bias correction: An N-dimensional probability density function transform for climate model simulations of multiple variables. *Climate Dynamics*, 50(1–2), 31–49. https://doi.org/10.1007/s00382-017-3580-6
- Comeau, L. E. L., Pietroniro, A., & Demuth, M. N. (2009). Glacier contribution to the north and South Saskatchewan rivers. *Hydrological Processes*, 23(18), 2640–2653. https://doi.org/10.1002/hyp
- De Vries, H., Kort, B., Teunis, B., Winters, M., & Beijk, V. (2021). Landelijk draaiboek waterverdeling en droogte (Tech. Rep.). Ministerie van Infrastructuur en Waterstaat. Retrieved from https://www.helpdeskwater.nl/onderwerpen/waterveiligheid/crisismanagement/ landelijk-draaiboek/
- Duethmann, D., Blöschl, G., & Parajka, J. (2020). Why does a conceptual hydrological model fail to predict discharge changes in response to climate change? *Hydrology and Earth System Sciences*, 1–28. https://doi.org/10.5194/hess-2019-652
- Ehsani, N., Vörösmarty, C. J., Fekete, B. M., & Stakhiv, E. Z. (2017). Reservoir operations under climate change: Storage capacity options to mitigate risk. Journal of Hydrology, 555, 435–446. https://doi.org/10.1016/j.jhydrol.2017.09.008
- Feyen, L., & Dankers, R. (2009). Impact of global warming on streamflow drought in Europe. Journal of Geophysical Research, 114, D17116. https://doi.org/10.1029/2008JD011438
- Finger, D., Vis, M., Huss, M., & Seibert, J. (2015). The value of multiple data set calibration versus model complexity for improving the performance of hydrological models in mountain catchments. Water Resources Research, 51, 1939–1958. https://doi.org/10.1002/2014WR015712
- Fischer, M., Huss, M., Barboux, C., & Hoelzle, M. (2014). The new Swiss Glacier Inventory SGI2010: Relevance of using high-resolution source data in areas dominated by very small glaciers. Arctic Antarctic and Alpine Research, 46(4), 933–945. https://doi.org/10.1657/1938-4246-46.4.933
- Fowler, K., Knoben, W., Peel, M., Peterson, T., Ryu, D., Saft, M., et al. (2020). Many commonly used rainfall-runoff models lack long, slow dynamics: Implications for runoff projections. *Water Resources Research*, 56, e2019WR025286. https://doi.org/10.1029/2019WR025286
- Frans, C., Istanbulluoglu, E., Lettenmaier, D. P., Fountain, A. G., & Riedel, J. (2018). Glacier recession and the response of summer streamflow in the Pacific Northwest United States, 1960–2099. Water Resources Research, 54, 6202–6225. https://doi.org/10.1029/2017WR021764
- Freudiger, D., Kohn, I., Seibert, J., Stahl, K., & Weiler, M. (2017). Snow redistribution for the hydrological modeling of alpine catchments. Wiley Interdisciplinary Reviews: Water, 4(5), e1232. https://doi.org/10.1002/wat2.1232
- Freudiger, D., Vis, M., & Seibert, J. (2020). *Quantifying the contributions to discharge of snow and glacier melt. Hydro-CH2018 project* (p. 49). Commissioned by the Federal Office for the Environment (FOEN).
- Fugger, S. (2018). Snow cover in the greater alpine region (2000–2017): Regional patterns, controls and change (Master thesis). BOKU University of Natural Resources and Life Sciences, Vienna, TUDelft University.
- Gerlinger, K., Moretti, G., Van Tiel, M., Kohn, I., Hänsler, A., Freudiger, D., et al. (2023). Hydrological model simulations of the Rhine basin and its tributaries: Streamflow and streamflow components. *Research Data*. https://doi.org/10.6094/UNIFR/233639
- Giq, A. (2014). Der frei fließende Rhein: Relevante Niedrig-und Mittelwasserstände (Tech. Rep.). Wasser-und Schifffahrtsverwaltung des Bundes. Retrieved from https://izw.baw.de/publikationen/vzb_dokumente_oeffentlich/0/Frei-fliessende-Rhein_2014.pdf
- Girons Lopez, M., Vis, M. J. P., Jenicek, M., Griessinger, N., & Seibert, J. (2020). Assessing the degree of detail of temperature-based snow routines for runoff modelling in mountainous areas in central Europe. *Hydrology and Earth System Sciences*, 24(9), 4441–4461. https://doi. org/10.5194/hess-24-4441-2020
- Gobiet, A., Kotlarski, S., Beniston, M., Heinrich, G., Rajczak, J., & Stoffel, M. (2014). 21st century climate change in the European Alps—A review. Science of the Total Environment, 493, 1138–1151. https://doi.org/10.1016/j.scitotenv.2013.07.050

- Godsey, S. E., Kirchner, J. W., & Tague, C. L. (2014). Effects of changes in winter snowpacks on summer low flows: Case studies in the Sierra Nevada, California, USA. *Hydrological Processes*, 28(19), 5048–5064. https://doi.org/10.1002/hyp.9943
- Gosling, S. N., Zaherpour, J., Mount, N. J., Hattermann, F. F., Dankers, R., Arheimer, B., & Breuer, L. (2017). A comparison of changes in river runoff from multiple global and catchment-scale hydrological models under global warming scenarios of 1°C, 2°C and 3°C. *Climatic Change*, 144, 577–595. https://doi.org/10.1007/s10584-016-1773-3
- Hänggi, P., & Weingartner, R. (2011). Variabilité interannuelle de l'écoulement et du climat dans le bassin du haut rhin 1808–2007. Hydrological Sciences Journal, 56(1), 34–50. https://doi.org/10.1080/02626667.2010.536549
- Hannaford, J., Lloyd-Hughes, B., Keef, C., Parry, S., & Prudhomme, C. (2011). Examining the large-scale spatial coherence of European drought using regional indicators of precipitation and streamflow deficit. *Hydrological Processes*, 25(7), 1146–1162. https://doi.org/10.1002/hyp.7725 Hausfather, Z., & Peters, G. P. (2020). Emissions—The 'business as usual' story is misleading. *Nature*, 577, 618–620. https://doi.org/10.1038/
- d4158,-020-00177-3 Hellwig, J., Stoelzle, M., & Stahl, K. (2021). Groundwater and baseflow drought responses to synthetic recharge stress tests. *Hydrology and Earth*
- System Sciences, 25(2), 1053–1068. https://doi.org/10.5194/hess-25-1053-2021 Hurkmans, R., Terink, W., Uijlenhoet, R., Torfs, P., Jacob, D., & Troch, P. A. (2010). Changes in streamflow dynamics in the Rhine basin under
- three high-resolution regional climate scenarios. Journal of Climate, 23(3), 679–699. https://doi.org/10.1175/2009JCL13066.1
- Huss, M. (2011). Present and future contribution of glacier storage change to runoff from macroscale drainage basins in Europe. Water Resources Research, 47, W07511. https://doi.org/10.1029/2010WR010299
- Huss, M., Bookhagen, B., Huggel, C., Jacobsen, D., Bradley, R. S., Clague, J. J., et al. (2017). Toward mountains without permanent snow and ice. *Earth's Future*, *5*, 418–435. https://doi.org/10.1002/2016EF000514
- Huss, M., & Farinotti, D. (2012). Distributed ice thickness and volume of all glaciers around the globe. *Journal of Geophysical Research*, 117, F04010. https://doi.org/10.1029/2012JF002523
- Huss, M., & Hock, R. (2018). Global-scale hydrological response to future glacier mass loss. *Nature Climate Change*, 8(2), 135–140. https://doi.org/10.1038/s41558-017-0049-x
- Huss, M., Jouvet, G., Farinotti, D., & Bauder, A. (2010). Future high-mountain hydrology: A new parameterization of glacier retreat. Hydrology and Earth System Sciences, 14(5), 815–829. https://doi.org/10.5194/hess-14-815-2010
- Immerzeel, W. W., Lutz, A. F., Andrade, M., Bahl, A., Biemans, H., Bolch, T., et al. (2020). Importance and vulnerability of the world's water towers. *Nature*, 577, 364–369. https://doi.org/10.1038/s41586-019-1822-y
- Jacob, D., Petersen, J., Eggert, B., Alias, A., Christensen, O. B., Bouwer, L. M., et al. (2014). EURO-CORDEX: New high-resolution climate change projections for European impact research. *Regional Environmental Change*, 14(2), 563–578. https://doi.org/10.1007/s10113-013-0499-2
- Jenicek, M., Seibert, J., Zappa, M., Staudinger, M., & Jonas, T. (2016). Importance of maximum snow accumulation for summer low flows in humid catchments. *Hydrology and Earth System Sciences*, 20(2), 859–874. https://doi.org/10.5194/hess-20-859-2016
- Jonkeren, O., Jourquin, B., & Rietveld, P. (2011). Modal-split effects of climate change: The effect of low water levels on the competitive position of inland waterway transport in the river Rhine area. Transportation Research Part A: Policy and Practice, 45(10), 1007–1019. https://doi. org/10.1016/j.tra.2009.01.004
- Jonkeren, O., Rietveld, P., & van Ommeren, J. (2007). Welfare effects of low water levels on the river Rhine through the inland waterway transport sector. *Journal of Transport Economics and Policy*, 41(3), 387–411.
- Jost, G., Moore, R. D., Menounos, B., & Wheate, R. (2012). Quantifying the contribution of glacier runoff to streamflow in the upper Columbia River Basin, Canada. *Hydrology and Earth System Sciences*, *16*(3), 849–860. https://doi.org/10.5194/hess-16-849-2012
- Junghans, N., Cullmann, J., & Huss, M. (2011). Evaluating the effect of snow and ice melt in an Alpine headwater catchment and further downstream in the River Rhine. *Hydrological Sciences Journal*, 56(6), 981–993. https://doi.org/10.1080/02626667.2011.595372
- Kaser, G., Großhauser, M., & Marzeion, B. (2010). Contribution potential of glaciers to water availability in different climate regimes. Proceedings of the National Academy of Sciences of the United States of America, 107(47), 20223–20227. https://doi.org/10.1073/pnas.1008162107
- Klein Tank, A. M. G., Wijngaard, J. B., Können, G. P., Böhm, R., Demarée, G., Gocheva, A., et al. (2002). Daily dataset of 20th-century surface air temperature and precipitation series for the European climate assessment. *International Journal of Climatology*, 22(12), 1441–1453. https:// doi.org/10.1002/joc.773
- Koboltschnig, G. R., Schöner, W., Zappa, M., & Holzmann, H. (2007). Contribution of glacier melt to stream runoff: If the climatically extreme summer of 2003 had happened in 1979. Annals of Glaciology, 46, 303–308. https://doi.org/10.3189/172756407782871260
- Kohn, I., Stahl, K., & Stölzle, M. (2019). Low flow events—A review in the context of climate change in Switzerland. Hydro-CH2018 Project. Albert-Ludwigs-Universität Freiburg. https://doi.org/10.6094/UNIFR/150448
- Konz, M., & Seibert, J. (2010). On the value of glacier mass balances for hydrological model calibration. *Journal of Hydrology*, 385(1–4), 238–246. https://doi.org/10.1016/j.jhydrol.2010.02.025
- Krajčí, P., Holko, L., Perdigão, R. A. P., & Parajka, J. (2014). Estimation of regional snowline elevation (RSLE) from MODIS images for seasonally snow covered mountain basins. *Journal of Hydrology*, 519, 1769–1778. https://doi.org/10.1016/j.jhydrol.2014.08.064
- Kundzewicz, Z. W., Radziejewski, M., & Pinskwar, I. (2006). Precipitation extremes in the changing climate of Europe. *Climate Research*, 31(1), 51–58. https://doi.org/10.3354/cr031051
- Ludwig, K., & Bremicker, M. (2006). The water balance model LARSIM: Design, content and applications (Tech. Rep.). Freiburg: Freiburger Schriften zur Hydrologie.
- Lutz, A. F., Immerzeel, W. W., Shrestha, A. B., & Bierkens, M. F. (2014). Consistent increase in High Asia's runoff due to increasing glacier melt and precipitation. *Nature Climate Change*, 4(7), 587–592. https://doi.org/10.1038/nclimate2237

Maisch, M. (2000). Die Gletscher der Schweizer Alpen: Gletscherhochstand 1850, aktuelle Vergletscherung, Gletscherschwund-Szenarien. Vdf, Hochschulverlag AG an der ETH.

- Marx, A., Kumar, R., Thober, S., Zink, M., Wanders, N., Wood, E., et al. (2017). Climate change alters low flows in Europe under a 1.5, 2, and 3 degree global warming. *Hydrology and Earth System Sciences Discussions*, 1–24. https://doi.org/10.5194/hess-2017-485
- Massarutto, A., Musolino, D., Pontoni, F., Carli, A. D., Senn, L., Paoli, L. D., et al. (2013). Analysis of historic events in terms of socio-economic and environmental impacts. Drought R&SPITech.Rep.No. 9. DROUGHT-R&SPI. Retrieved from http://www.isa.ulisboa.pt/ceabn/uploads/ docs/projectos/drought/DROUGHT_TR_9.pdf
- McCarthy, M., Meier, F., Fatichi, S., Stocker, B. D., Shaw, T. E., Miles, E., et al. (2022). Glacier contributions to river discharge during the current Chilean Megadrought. *Earth's Future*, 10, e2022EF002852. https://doi.org/10.1029/2022EF002852
- MeteoSwiss. (2017). Documentation of MeteoSwiss grid-data products daily mean, minimum and maximum temperature: TabsD, TminD, TmaxD (pp. 1–4).
- MeteoSwiss. (2019). Documentation of MeteoSwiss grid-data products daily precipitation (final analysis): RhiresD (pp. 1-4).

Meyer, J., Kohn, I., Stahl, K., Hakala, K., Seibert, J., & Cannon, A. J. (2019). Effects of univariate and multivariate bias correction on hydrological impact projections in alpine catchments. *Hydrology and Earth System Sciences*, 23(3), 1339–1354. https://doi.org/10.5194/hess-23-1339-2019 Mosley, L. M. (2015). Drought impacts on the water quality of freshwater systems; review and integration. *Earth-Science Reviews*, 140, 203–214. https://doi.org/10.1016/j.earscirev.2014.11.010

Müller, F., Catfish, T., & Müller, G. (1977). Firn und eis der Schweizer alpen. Norsk Geografisk Tidsskrift-Norwegian. Journal of Geography, 31(2), 87–88. https://doi.org/10.1080/00291957708545328

- Nepal, S. (2016). Impacts of climate change on the hydrological regime of the Koshi river basin in the Himalayan region. Journal of Hydro-environment Research, 10, 76–89. https://doi.org/10.1016/j.jher.2015.12.001
- Nie, Y., Pritchard, H. D., Liu, Q., Hennig, T., Wang, W., Wang, X., et al. (2021). Glacial change and hydrological implications in the Himalaya and Karakoram. *Nature Reviews Earth & Environment*, 2(2), 91–106. https://doi.org/10.1038/s43017-020-00124-w
- Paul, F., Frey, H., & Bris, R. L. (2011). A new glacier inventory for the European Alps from Landsat TM scenes of 2003: Challenges and results. Annals of Glaciology, 52(59), 144–152. https://doi.org/10.3189/172756411799096295
- Pelto, M. S., Dryak, M., Pelto, J., Matthews, T., & Perry, L. B. (2022). Contribution of glacier runoff during heat waves in the Nooksack river basin USA. Water, 14(7), 1145. https://doi.org/10.3390/w14071145
- Pfister, C., Weingartner, R., & Luterbacher, J. (2006). Hydrological winter droughts over the last 450 years in the upper Rhine basin: A methodological approach. *Hydrological Sciences Journal*, 51(5), 966–985. https://doi.org/10.1623/hysj.51.5.966
- Pritchard, H. D. (2019). Asia's shrinking glaciers protect large populations from drought stress. *Nature*, 569, 649–654. https://doi.org/10.1038/ s41586-019-1240-1
- Rasmijn, L. M., Van Der Schrier, G., Bintanja, R., Barkmeijer, J., Sterl, A., & Hazeleger, W. (2018). Future equivalent of 2010 Russian heatwave intensified by weakening soil moisture constraints. *Nature Climate Change*, 8(5), 381–385. https://doi.org/10.1038/s41558-018-0114-0
- Rauthe, M., Steiner, H., Riediger, U., Mazurkiewicz, A., & Gratzki, A. (2013). A Central European precipitation climatology—Part I: Generation and validation of a high-resolution gridded daily data set (HYRAS). *Meteorologische Zeitschrift*, 22(3), 235–256. https://doi. org/10.1127/0941-2948/2013/0436
- Razafimaharo, C., Krähenmann, S., Höpp, S., Rauthe, M., & Deutschländer, T. (2020). New high-resolution gridded dataset of daily mean, minimum, and maximum temperature and relative humidity for Central Europe (HYRAS). *Theoretical and Applied Clinatology*, 142(3), 1531–1553. https://doi.org/10.1007/s00704-020-03388-w
- Rebetez, M., Mayer, H., Dupont, O., Schindler, D., Gartner, K., Kropp, J. P., & Menzel, A. (2006). Heat and drought 2003 in Europe: A climate synthesis. Annals of Forest Science, 63(6), 569–577. https://doi.org/10.1051/forest:2006043
- Rottler, E., Bronstert, A., Bürger, G., & Rakovec, O. (2021). Projected changes in Rhine River flood seasonality under global warming. Hydrology and Earth System Sciences, 25(5), 2353–2371. https://doi.org/10.5194/hess-25-2353-2021
- Samaniego, L., Thober, S., Kumar, R., Wanders, N., Rakovec, O., Pan, M., et al. (2018). Anthropogenic warming exacerbates European soil moisture droughts. *Nature Climate Change*, 8(5), 421–426. https://doi.org/10.1038/s41558-018-0138-5
- Schmucki, E., Marty, C., Fierz, C., & Lehning, M. (2015). Simulations of 21st century snow response to climate change in Switzerland from a set of RCMs. International Journal of Climatology, 35(11), 3262–3273. https://doi.org/10.1002/joc.4205
- Schnorbus, M., Werner, A., & Bennett, K. (2014). Impacts of climate change in three hydrologic regimes in British Columbia, Canada. Hydrological Processes, 28(3), 1170–1189. https://doi.org/10.1002/hyp.9661
- Seibert, J., Vis, M., Kohn, I., Weiler, M., & Stahl, K. (2018). Technical note: Representing glacier geometry changes in a semi-distributed hydrological model. *Hydrology and Earth System Sciences*, 22(4), 2211–2224. https://doi.org/10.5194/hess-22-2211-2018
- Seibert, J., & Vis, M. J. (2012). Teaching hydrological modeling with a user-friendly catchment-runoff-model software package. *Hydrology and Earth System Sciences*, 16(9), 3315–3325. https://doi.org/10.5194/hess-16-3315-2012
- Shepherd, T. G., Boyd, E., Calel, R. A., Chapman, S. C., Dessai, S., Dima-west, I. M., et al. (2018). Storylines: An alternative approach to representing uncertainty in physical aspects of climate change. *Climate Change*, 151, 555–571. https://doi.org/10.1007/s10584-018-2317-9
- Shrestha, R. R., Schnorbus, M. A., Werner, A. T., & Berland, A. J. (2012). Modelling spatial and temporal variability of hydrologic impacts of climate change in the Fraser River basin, British Columbia, Canada. *Hydrological Processes*, 26(12), 1840–1860. https://doi.org/10.1002/ hyp.9283
- Sillmann, J., Shepherd, T. G., van den Hurk, B., Hazeleger, W., Martius, O., Slingo, J., & Zscheischler, J. (2021). Event-based storylines to address climate risk. *Earth's Future*, 9, e2020EF001783. https://doi.org/10.1029/2020EF001783
- Sommer, C., Malz, P., Seehaus, T. C., Lippl, S., Zemp, M., & Braun, M. H. (2020). Rapid glacier retreat and downwasting throughout the European Alps in the early 21st century. *Nature Communications*, 11(1), 3209. https://doi.org/10.1038/s41467-020-16818-0
- Sorg, A., Bolch, T., Stoffel, M., Solomina, O., & Beniston, M. (2012). Climate change impacts on glaciers and runoff in Tien Shan (Central Asia). *Nature Climate Change*, 2(10), 725–731. https://doi.org/10.1038/nclimate1592
- Stahl, K., Kohn, I., Blauhut, V., Urquijo, J., De Stefano, L., Acácio, V., et al. (2016). Impacts of European drought events: Insights from an international database of text-based reports. *Natural Hazards and Earth System Sciences*, 16(3), 801–819. https://doi.org/10.5194/nhess-16-801-2016
- Stahl, K., Weiler, M., Freudiger, D., Kohn, I., Seibert, J., Vis, M., et al. (2017). The snow and glacier melt components of streamflow of the river Rhine and its tributaries considering the influence of climate change. Final report to the International Commission for the Hydrology of the Rhine (CHR) (Tech. Rep.). Retrieved from https://www.chr-khr.org/sites/default/files/chrpublications/asg-rhine-final-report-2017_0.pdf
- Stahl, K., Weiler, M., Kohn, I., Freudiger, D., Seibert, J., Vis, M., et al. (2016). *The snow and glacier melt components of streamflow of the river Rhine and its tributaries considering the influence of climate change (Synthesis report)*. International Commission for the Hydrology of the Rhine Basin. Retrieved from https://www.chr-khr.org/sites/default/files/chrpublications/asg-rhein_synthesis_en.pdf
- Stahl, K., Weiler, M., van Tiel, M., Kohn, I., Hänsler, A., Freudiger, D., et al. (2022). Impact of climate change on the rain, snow and glacier melt components of streamflow of the river Rhine and its tributaries (Tech. Rep. No. CHR Rep. No. 128). Lelystad: International Commission for the Hydrology of the Rhine Basin. Retrieved from https://www.chr-khr.org/sites/default/files/ASG-II_Synthese_EN_mit-Links.pdf
- Stoelzle, M., Schuetz, T., Weiler, M., Stahl, K., & Tallaksen, L. M. (2020). Beyond binary baseflow separation: A delayed-flow index for multiple streamflow contributions. *Hydrology and Earth System Sciences*, 24(2), 849–867. https://doi.org/10.5194/hess-24-849-2020
- Tague, C., & Grant, G. E. (2009). Groundwater dynamics mediate low-flow response to global warming in snow-dominated alpine regions. Water Resources Research, 45, W07421. https://doi.org/10.1029/2008WR007179
- Teuling, A. J. (2018). A hot future for European droughts. *Nature Climate Change*, 8(5), 364–365. https://doi.org/10.1038/s41558-018-0154-5 Udall, B., & Overpeck, J. (2017). The twenty-first century Colorado River hot drought and implications for the future. *Water Resources Research*, 53, 2404–2418. https://doi.org/10.1111/j.1752-1688.1969.tb04897.x
- Van der Wiel, K., Lenderink, G., & De Vries, H. (2021). Physical storylines of future European drought events like 2018 based on ensemble climate modelling. Weather and Climate Extremes, 33, 100350. https://doi.org/10.1016/j.wace.2021.100350

- Van Slobbe, E., Werners, S. E., Riquelme-Solar, M., Bölscher, T., & van Vliet, M. T. (2016). The future of the Rhine: Stranded ships and no more salmon? *Regional Environmental Change*, 16(1), 31–41. https://doi.org/10.1007/s10113-014-0683-z
- Van Tiel, M., Freudiger, D., Kohn, I., Seibert, J., Weiler, M., & Stahl, K. (2022a). Hydrological modelling of the glacierized headwater catchments in the Rhine basin: Streamflow, streamflow components and glacier area model simulations. *Research Data*. https://doi.org/10.6094/ UNIFR/226494
- Van Tiel, M., Freudiger, D., Kohn, I., Seibert, J., Weiler, M., & Stahl, K. (2022b). Hydrological modelling of the glacierized headwater catchments in the Rhine Basin: Technical Report. Albert-Ludwigs-Universität Freiburg. https://doi.org/10.6094/UNIFR/226492
- Van Tiel, M., Teuling, A. J., Wanders, N., Vis, M. J., Stahl, K., & Van Loon, A. F. (2018). The role of glacier changes and threshold definition in the characterisation of future streamflow droughts in glacierised catchments. *Hydrology and Earth System Sciences*, 22(1), 463–485. https:// doi.org/10.5194/hess-22-463-2018
- Van Tiel, M., Van Loon, A. F., Seibert, J., & Stahl, K. (2021). Hydrological response to warm and dry weather: Do glaciers compensate? Hydrology and Earth System Sciences, 25(6), 3245–3265. https://doi.org/10.5194/hess-25-3245-2021
- Van Tiel, M., Weiler, M., Freudiger, D., Moretti, G., Kohn, I., Gerlinger, K., & Stahl, K. (2023). Stress-test storyline simulations of the Rhine basin: Streamflow and streamflow components. *Research Data*. https://doi.org/10.6094/UNIFR/233644
- Vinke, F., van Koningsveld, M., van Dorsser, C., Baart, F., van Gelder, P., & Vellinga, T. (2022). Cascading effects of sustained low water on inland shipping. *Climate Risk Management*, 35, 100400. https://doi.org/10.1016/j.crm.2022.100400
- Viviroli, D., & Weingartner, R. (2004). The hydrological significance of mountains: From regional to global scale. Hydrology and Earth System Sciences, 8(6), 1017–1030. https://doi.org/10.5194/hess-8-1017-2004
- Wan, W., Zhao, J., Popat, E., Herbert, C., & Döll, P. (2021). Analyzing the impact of streamflow drought on hydroelectricity production: A global-scale study. Water Resources Research, 57, e2020WR028087. https://doi.org/10.1029/2020WR028087
- Weiler, M., Seibert, J., & Stahl, K. (2018). Magic components—Why quantifying rain, snowmelt, and icemelt in river discharge is not easy. *Hydrological Processes*, 32(1), 160–166. https://doi.org/10.1002/hyp.11361
- Yevenes, M. A., Figueroa, R., & Parra, O. (2018). Seasonal drought effects on the water quality of the Biobío River, Central Chile. Environmental Science and Pollution Research, 25(14), 13844–13856. https://doi.org/10.1007/s11356-018-1415-6
- Zappa, M., & Kan, C. (2007). Extreme heat and runoff extremes in the Swiss Alps. *Natural Hazards and Earth System Science*, 7(3), 375–389. https://doi.org/10.5194/nhess-7-375-2007
- Zekollari, H., Huss, M., & Farinotti, D. (2019). Modelling the future evolution of glaciers in the European Alps under the EURO-CORDEX RCM ensemble. *The Cryosphere*, 13(4), 1125–1146. https://doi.org/10.5194/tc-13-1125-2019